Dear Editor and Reviewers:

We greatly appreciate your review of our manuscript and the valuable suggestions provided. Please find below responses to reviewer comments, along with a description of the revisions made to the manuscript.

Response to Reviewer 1:
Thank you for your suggestion. We have used abbreviations after their spelling out in the revised manuscript.

Response to Reviewer 2:
We greatly appreciate your review of our manuscript.

Response to Reviewer 3:
The revised manuscript has been carefully reviewed by experienced editors whose first language is English.

Response to Reviewer 4:
We greatly appreciate your review of our manuscript.

Response to Reviewer 5:
The revised manuscript has been carefully reviewed and polished by experienced editors whose first language is English.